Author's response to reviews

Title: The morbidity and mortality following a diagnosis of peripheral arterial disease: Long term follow-up of a large database

Authors:
Jaime Caro (jcaro@caroresearch.com)
Kristen Migliaccio-Walle (kmw@caroresearch.com)
Khajak J Ishak (jishak@caroresearch.com)
Irina Proskorovsky (iproskorovsky@caroresearch.com)

Version: 2 Date: 7 March 2005

Author's response to reviews:
March 7, 2005

Peter Newmark
Editor-in- Chief
BMC Cardiovascular Disorders
BioMed Central

Manuscript Title: The morbidity and mortality following a diagnosis of peripheral arterial disease: Long-term follow-up of a large database.

Manuscript ID: 2077431325629004

Revision Description: Two cross references, (pages 8, 9, & 11) to the figures did not update properly in the original version of the manuscript that was uploaded for review. In that version, they read "Error! Reference source not found." They have been corrected in this version.

Sincerely,

J. Jaime Caro, MDCM,FRCPC,FACP